Author’s response to reviews

Title: A cluster randomised controlled trial of an occupational therapy intervention for residents with stroke living in UK care homes (OTCH): Study protocol

Authors:

Cath M. Sackley (c.sackley@uea.ac.uk)
Chris R. Burton (c.burton@bangor.ac.uk)
Sandy Herron-Marx (sandy.herron-marx@warwick.ac.uk)
Karen Lett (k.l.lett@bham.ac.uk)
Jonathan Mant (jm677@medschl.cam.ac.uk)
Andrea K. Roalfe (a.k.roalfe@bham.ac.uk)
Leslie J. Sharp (lesliejohnsharp@gmail.com)
Bart Sheehan (b.sheehan@warwick.ac.uk)
Katie E. Stant (k.e.stant@bham.ac.uk)
Marion F. Walker (marion.walker@nottingham.ac.uk)
Caroline L. Watkins (clwatkins@uclan.ac.uk)
Keith Wheatley (k.wheatley@bham.ac.uk)
Jane Williams (jane.williams@porthosp.nhs.uk)
Guiqing L. Yao (g.l.yao@bham.ac.uk)
Max G. Feltham (m.g.feltham@bham.ac.uk)

Version: 2 Date: 27 April 2012

Author’s response to reviews: see over
A cluster randomised controlled trial of an occupational therapy intervention for residents with stroke living in UK care homes (OTCH): Study protocol


BMC Neurology

Dear Editor,

Thank you for inviting us to revise our manuscript to address the concerns raised by your Senior Medical Editors. Attached is the amended manuscript, where I have highlighted the changes from the original submission (i.e., discussion of a previous study published in the BMJ and email address of author 7). We certainly appreciate these comments and hope that you find our answers satisfactory. Additionally, I have attached the requested documentation, which includes:

- the journal paper from our reference list you requested,
- proof of ethics approval for the study,
- proof of funding for the project.

We very much look forward to hearing from you in due course and remain at your disposal to respond to any further questions and clarifications should this be deemed necessary.

Yours faithfully,

Max Feltham
Research Fellow
University of Birmingham